Author's response to reviews

Title: Comparative Effectiveness of Monotherapies and Combination Therapies for Patients with Hypertension: Protocol for a Systematic Review with Network Meta-Analyses

Authors:

Brian Hutton (bhutton@ohri.ca)
Jennifer Tetzlaff (jitetzlaff@ohri.ca)
Fatemah Yazdi (fayazdi@ohri.ca)
Justin Thielman (jthielmann@ohri.ca)
Frans H.H. Leenen (frans.leenen@ottawaheart.ca)
Salmaan Kanji (skanji@ottawahospital.on.ca)
Dean Fergusson (dafergusson@ohri.ca)
Lise Bjerre (lbjerre@bruyere.org)
Ed Mills (edward.mills@uottawa.ca)
Kristian Thorlund (thorluk@univmail.cis.mcmaster.ca)
Andrea Tricco (triccoa@smh.ca)
Sharon Straus (sharon.straus@utoronto.ca)
David Moher (dmoher@ohri.ca)

Version: 2 Date: 8 April 2013

Author's response to reviews: see over
April 5, 2013

Systematic Reviews
c/o BioMed Central
236 Gary’s Inn Road
London, WC1X 8HB
United Kingdom

Dear Editors-in-Chief:

Please find enclosed our manuscript entitled ‘Comparative Effectiveness of Monotherapies and Combination Therapies for Patients with Hypertension: Protocol for a Systematic Review with Network Meta-Analyses’ for consideration of publication with the Systematic Reviews journal.

Our systematic review explores the relative benefits and safety of different pharmacotherapies for management of non-diabetic patients with hypertension, whether of a mild or moderate to severe nature. The described review has been funded in Canada by the Canadian Institutes of Health Research and the Drug Safety and Effectiveness Network, where it underwent a peer review process previously. We appreciate your consideration to publish our systematic review in the Systematic Reviews journal.

Should you have any questions regarding this manuscript, please don’t hesitate to contact me.

Sincerely,

Dr. Brian Hutton, PhD
Ottawa Hospital Research Institute
Center for Practice Changing Research
501 Smyth Road, Ottawa, Ontario, K1H 8L6
Phone: 613-737-8899 (ext. 73827)
email bhutton@ohri.ca